Summary of May 12 Meeting to Discuss Plans and Schedule Outlook for Commissioning Cryogenics Aspects of the HINS    325 MHz Cavity Test Cave

Cryo studies are completed. Studies were done with and without a heat load (up to 50W), and with and without HTS running. Pressure is about 5 psig “full flow” (with HTS running) and 4 psig with HTS off. Pressure variation in all cases is approximately 0.3 psi peak to peak. Pressure does not seem to vary any more when there is a heat load. We are now warming up, and will be warm by tomorrow (Thursday) afternoon.
Elmie reported that Fred Lewis needs 1 day to complete his wiring. This will be done on Friday or Monday.

It was agreed that a rupture disk is needed on the cavity vacuum system (present in the system when the turbo is valved out).

Next week, Alex will move the clean room (to be used in the future for the coupler for pulsed tests) into the cave, and then put the roof on and sandbags in. Once this is done, the cavity can be moved to meson and installed into the cryostat. The move will likely happen the week of May 24. Bob pointed out the need to clearly review the cavity handling procedure with (and solicit feedback from) all of those involved before actually attempting to move it. The actual moving and installation will be done by Henry and Phil.
Alex is waiting for the ion pump for the cavity. That is about 3 weeks away, so once the cavity is installed, instrumentation work can proceed. (temperature sensor installation – arranged by Timergali, Xray detector installation – Aseet and Bob, solenoid study apparatus – Yuri/Timergali etc.)

Vacuum work will be done by superconducting cavity experts, or by Henry and Phil plus superconducting cavity experts, for training.

Bob Wagner pointed out that before the helium vessel pressure test, we should verify that the cavity is indeed filled with 50mbar psig (or something near atmospheric pressure, as opposed to vacuum) so as to avoid deforming the cavity. This can be done by measuring the tune.

